Author's response to reviews

Title: Haiti and the health marketplace: The role of the private, informal market in filling the gaps left by the state

Authors:

Jo Durham (durhamjo@yahoo.com)
Markus J Michael (marcos.j.michael@gmail.com)
Peter S Hill (p.hill@sph.uq.edu.au)
Enrico Pavignani (enricopavignani@hotmail.com)

Version: 8 Date: 10 September 2015

Author's response to reviews: see over
10th September 2015

Dear Editor

Thank you for the acceptance of the manuscript *Haïti and the health marketplace: The role of the private, informal market in filling the gaps left by the state*

We have:

1) Please remove all the additional files (the ethics documentation is not for publication).
   We have removed the additional files (e.g. ethics application and approval)

2) Please include the title page into the main manuscript.
   We have included the title page into the main manuscript

3) Competing interests (before the references)
   Included competing interests as follows: 'The author(s) declare that they have no competing interests

4) Authors’ contributions (before the references)
   We have added:
   JD contributed to the analysis and synthesised the data from the case study and prepared the initial manuscript. EP, MM and PH designed the study and undertook the field research and initial analysis. All authors contributed to the manuscript. All authors read and approved the final manuscript.

5) Acknowledgements (before the references)
   We have added:
   This research was partly funded by Ministry of Foreign Affairs of Denmark
   The funding body had no role in the design, design, in the collection, analysis, and interpretation of data; in the writing of the manuscript; and in the decision to submit the manuscript for publication.

We feel the paper fills a critical gap in our understanding of health systems in fragile states and fits the interest of you readership and we look forward to its publication.

Yours sincerely

Dr Jo Durham
University of Queensland
School of Public Health